Author's response to reviews

Title: Methylthioadenosine (MTA) inhibits melanoma cell proliferation and in vivo tumor growth.

Authors:

Pedro Andreu-Perez (pandreu@ir.vhebron.net)
Javier Hernandez-Losa (j hernandez@vheron.net)
Teresa Moline (tmolne@vhebron.net)
Anna Pujol (apujol@ir.vhebro.net)
Javier Cortes (jacortes@vhebron.net)
Judit Grueso (jgrueso@ir.vhebron.net)
Rosa Gil (rogil@ir.vhebron.net)
Matias A Avila (maavila@unav.es)
Juan A Recio (jarecio@ir.vhebron.net)

Version: 4 Date: 31 May 2010

Author's response to reviews:

Dear Editor:

Thank you for accepting the manuscript for publication in BMC cancer.

In this final version we have added the minor essential revisions suggested by Referee 1 and the major formatting revisions suggested by the editorial production team. We have checked the author list and corrected the author’s contributions according the example provided. We also made the changes related with the additional files.

Please, contact us if you have any problems and questions

Best wishes,

Fdo.

Juan A. Recio Conde